Author's response to reviews

Title: Statins in Candidemia: clinical outcomes from a matched cohort study

Authors:

Graeme N Forrest (forrestg@ohsu.edu)
Angela M Kopack (akopack@gmail.com)
Eli N Perencevich (eli-perencevich@uiowa.edu)

Version: 5 Date: 2 June 2010

Author's response to reviews:

Everything looks good. Just minor changes on the figure

Minor Changes:

1) Added Footnote to Figure legend stating ICU =Intensive care unit in main manuscript
2) On Figure 1 - added "for matching" to lower right box on non statin side.

Thanks